Author’s response to reviews

Title: Lethal thrombosis of the iliac artery caused by Aspergillus fumigatus after liver transplantation. Case report and review of the literature.

Authors:

Jan-Paul Gundlach (Jan-Paul.Gundlach@uksh.de)
Rainer Günther (Rainer.Guenther@uksh.de)
Helmut Fickenscher (Helmut.Fickenscher@uksh.de)
Marcus Both (Marcus.Both@uksh.de)
Christoph Röcken (Christoph.Roecken@uksh.de)
Thomas Becker (Thomas.Becker@uksh.de)
Felix Braun (Felix.Braun@uksh.de)

Version: 1 Date: 16 Sep 2019

Author’s response to reviews:

Dear Editor,

thank you very much for your quick response with thoughtful comments. We adopted or manuscript accordingly and included it in the revised version. In detail, we erased the year of birth as well as the date of diagnosis in the manuscript. In addition, we added in the declaration section that the patient had signed informed consent before transplantation including the consent to publication.

We think that this clarifies the legal status and we hope that these modifications will completely satisfy you.

Yours sincerely, Jan-Paul Gundlach